Author’s response to reviews

Title: The lack of public health research output from India

Authors:

- Lalit Dandona (dandona@asci.org.in)
- Yegnanarayana S Sivan (yssivan@rediffmail.com)
- Mukkamala N Jyothi (jyothimukkamala@rediffmail.com)
- V S Udaya Bhaskar (uday@asci.org.in)
- Rakhi Dandona (rakhi@asci.org.in)

Version: 6 Date: 16 November 2004

Author’s response to reviews:

16 November 2004

We have now made all the formatting changes that were advised:

1. Date removed from title page.
2. Figure/Table placement text made unbold.
4. The tables are changed to portrait format.
5. Figure title/number removed from image page.
6. Figure cropped.

Thank you very much.

Best regards,

Lalit Dandona